Author’s response to reviews

Title: Health Care Professionals' Attitudes towards Population-Based Genetic Testing and Risk-Stratification for Ovarian Cancer: A Cross-sectional Survey

Authors:

Katie Hann (k.hann@ucl.ac.uk)
Lindsay Fraser (l.fraser@ucl.ac.uk)
Lucy Side (l.side@ucl.ac.uk)
Sue Gessler (s.gessler@ucl.ac.uk)
Jo Waller (j.waller@ucl.ac.uk)
Saskia Sanderson (saskia.sanderson@ucl.ac.uk)
Madeleine Freeman (maddie.freeman@ucl.ac.uk)
Ian Jacobs (i.jacobs@unsw.edu.au)
Anne Lanceley (a.lanceley@ucl.ac.uk)

Version: 1 Date: 05 Jul 2017

Author’s response to reviews:

Dear Ms Nourollahi,

Thank you for considering our manuscript for review.

We have revised our ethics statement to clarify how participants consented to take part in the study and we have included a completed STROBE checklist in the supplementary files. The section entitled 'Supplementary files' can be found at the end of the manuscript, after the list of abbreviations.

We look forward to hearing from you.

Yours sincerely,

Katie Hann